Author's response to reviews

Title: Conceptualization of Common LR: A useful Tool for Clinical Diagnostic Reasoning

Authors:

Hamideh Moosapour (dr_moosapour@yahoo.com)
Mohsin Raza (Mohsinreza60@yahoo.com)
Akbar Soltani (soltania@tums.ac.ir)

Version: 3 Date: 12 June 2010

Author's response to reviews: see over
Dear editor
According to reviewers' comments, new version of article has been changed a lot. Thanks for your cooperation and offering more time to do revisions. We hope that all the changes that were required are made. Kindly let us know if more changes are necessary.

Respectfully yours,

Akbar Soltani, MD
Endocrinology & Metabolism Research Centre, Fifth floor, Shariati Hospital, Northern Kargar Ave, Tehran 14114, Iran
Tel: +98 21 84902476